Author’s response to reviews

Title: What can global health institutions do to help strengthen health systems in low income countries?

Authors:

Dina Balabanova (dina.balabanova@lshtm.ac.uk)
Martin McKee (martin.mckee@lshtm.ac.uk)
Anne Mills (anne.mills@lshtm.ac.uk)
Gill Walt (gill.walt@lshtm.ac.uk)
Andy Haines (andy.haines@lshtm.ac.uk)

Version: 3 Date: 18 June 2010

Author’s response to reviews:

We have complied with all the requests relating to formatting. Specifically, 2 boxes have been moved into the text and the 3rd has been redrawn as a table. Shading has been removed from the other table. The link that did not work has been updated (the Bamako website has been taken down but the document is on the WHO site) etc.

The one reference in press is still in press. We have checked the text in detail.